\begin{table}%t6[!htbp]
\caption{Summary of the line identifications for \ion{Fe}{xxiii} 
The columns indicate: 
1)~the  indices corresponding to Tables~\ref{tab:levels};
2,3)~the relative intensities (in photons, scaled to the 1$-$15
10.980~\AA\ line and calculated at log $T[{\rm K}]=7.2$) 
typical of astrophysical and laser plasmas (10$^{12}$, 10$^{19}$~\cmt);
4)~the  wavelengths calculated from our best energies $E_{\rm best}$;
5)~observed  wavelengths $\lambda_{\rm observed}$;
 some blends are indicated (bl~= blend; bl-mr~= blend in medium-resolution spectra;
bl-w~= blend with a weak line); 
lines with no or a tentative identification have a question mark;
%
6,7)~previous identifications consistent or not with ours 
(with  observed  wavelengths in \AA; note that 
 observed and calculated wavelengths in the cited literature
can differ from the values reported here).
Legenda: 
%
Bo78: \cite{boiko_etal:78} (see also original work cited in this
review paper);
B78: \cite{bromage_etal:78};
B85: \cite{burkhalter_etal:85};
B02: \cite{brown_etal:02};
D72: \cite{doschek_etal:72};
F79: \cite{fawcett_etal:79};
F00: \cite{feldman_etal:00};
%F84: \cite{finkenthal_etal:84};
H80: \cite{hinnov_suckewer:80};
K74: \cite{kastner_etal:74};
LP80: \cite{lawson_peacock:80};
S80: \cite{spector_etal:80}
SF86: \cite{seely_feldman:86};
SR95: \cite{sugar_rowan:95};
W75: \cite{widing:75};
W98: \cite{wargelin_etal:98}.
The final column indicates weather the lines have only been 
observed in high-density plasmas (H).
}
\label{tab:id}
%\centering
\par
%\scriptsize
\footnotesize
\begin{tabular}[c]{@{}ccclllll@{}}
\hline\hline\noalign{\smallskip}
 $i{-}j$   &  $Int$ &  $Int$ & $\lambda_{\rm best}$(\AA) & $\lambda_{\rm observed}$(\AA) & Same ID & Diff. ID & \\ 
   &  10$^{12}$ &  10$^{19}$ &  & & & & \\ 
\hline\noalign{\smallskip}
%
% fe_23_emissn_ident.tex
%
%Calculated: 15/Aug/04 17:35:07
%cutoff=1E-05 cutoff=0.012 cutoff=0.0035 cutoff=0.006 cutoff=0.006
%\begin{longtable}{@{}ccclllll@{}}
%\caption{Line details. The relative intensities  (photons)  $Int=N_{j} A_{ji}/N_{\rm e}$ are normalised to 1.1e-11} \label{tab:lines} \\
%\noalign{\smallskip} \hline
% $i{-}j$   &  $Int$ &  $Int$ & $\lambda_{\rm best}$(\AA) & $\lambda_{\rm observed}$(\AA) & Same ID & Diff. ID & \\ 
%   & 1.0 10$^{12}$ & 1.0 10$^{19}$ &  & & & & \\ 
\par
%2-64 & 6.7 10$^{-4}$ & 7.7 10$^{-3}$ & 8.270 $\pm$ 0.004 &      &     &      &   \\ 
4--78 & 4.7 $\times$ 10$^{-4}$ & 1.4 $\times$ 10$^{-2}$ & 8.273 $\pm$ 0.002 & 8.273 $\pm$ 0.002 Bo78 (bl) &  B78, F79  &     &  H  \\
4--79 & 2.0 $\times$ 10$^{-4}$ & 1.1 $\times$ 10$^{-2}$ & 8.273 $\pm$ 0.002 & ''                               &   &     &  H  \\
%4--80 & 7.8 10$^{-4}$ & 6.5 10$^{-3}$ & 8.273 $\pm$ 0.003 &      &     &      &   \\ 
3--65 & 4.3 $\times$ 10$^{-4}$ & 1.4 $\times$ 10$^{-2}$ & 8.290 $\pm$ 0.003 & ? bl Fe~XXIV 8.285  $\pm$ 0.002 Bo78   &   &     & H   \\
1--52 & 0.17 & 4.1 $\times$ 10$^{-2}$ & 8.303 $\pm$ 0.003 & 8.3038 $\pm$ 0.0003 W98 (bl Fe~XXIV 8.316)   & D72, B78  &     &    \\
1--50 & 2.4 $\times$ 10$^{-2}$ & 1.6 $\times$ 10$^{-2}$ & 8.315 $\pm$ 0.002 & 8.315 $\pm$ 0.002 Bo78  (bl Fe~XXIV)  &     &     &    \\   % Int=3.40e-02
5--81 & 3.1 $\times$ 10$^{-4}$ & 1.2 $\times$ 10$^{-2}$ & 8.454 $\pm$ 0.004 & ? 8.452 $\pm$ 0.002  Bo78    &     &     &  H  \\
%5--77 & 1.0 10$^{-3}$ & 6.1 10$^{-3}$ & 8.478 $\pm$ 0.004 &      &     &      &   \\ 
2--53 & 9.6 $\times$ 10$^{-3}$ & 2.9 $\times$ 10$^{-2}$ & 8.529 $\pm$ 0.002 & 8.529 $\pm$ 0.002 Bo78,W98 (bl ?)  &  B78  &     &    \\
3--54 & 1.8 $\times$ 10$^{-2}$ & 6.1 $\times$ 10$^{-2}$ & 8.550 $\pm$ 0.001 & 8.550 $\pm$ 0.002 Bo78  (8.546 W98)  &  B78  &   &   \\
3--53 & 7.0 $\times$ 10$^{-3}$ & 2.1 $\times$ 10$^{-2}$ & 8.551 $\pm$ 0.001 & '' ~(weak)     &     &     &    \\
4--55 & 3.1 $\times$ 10$^{-2}$ & 9.9 $\times$ 10$^{-2}$ & 8.614 $\pm$ 0.001 & 8.614 $\pm$ 0.002 Bo78 (8.617 W98)  &   B78 &     &  \\  % Int=3.35e-02
4--54 & 5.8 $\times$ 10$^{-3}$ & 2.0 $\times$ 10$^{-2}$ & 8.618 $\pm$ 0.001 & ''  ~(bl)  &     &     &  H  \\  % Int=6.38e-03
7--87 & 3.1 $\times$ 10$^{-4}$ & 9.2 $\times$ 10$^{-3}$ & 8.630 $\pm$ 0.003 & 8.630 $\pm$ 0.002 Bo78    &   B78  &     &  H  \\
7--83 & 1.4 $\times$ 10$^{-4}$ & 8.1 $\times$ 10$^{-3}$ & 8.643 $\pm$ 0.003 & 8.643 $\pm$ 0.002  Bo78    &  B78   &     &  H  \\
8--86 & 2.3 $\times$ 10$^{-4}$ & 1.1 $\times$ 10$^{-2}$ & 8.664 $\pm$ 0.003 & 8.664 $\pm$ 0.002 Bo78    &  B78   &     &  H  \\
6--70 & 4.9 $\times$ 10$^{-3}$ & 1.8 $\times$ 10$^{-2}$ & 8.672 $\pm$ 0.003 & ''   &  (B78)   &     &  H  \\
8--84 & 3.6 $\times$ 10$^{-4}$ & 3.2 $\times$ 10$^{-2}$ & 8.672 $\pm$ 0.003 & 8.672 $\pm$ 0.002 Bo78    &  B78   &     &  H  \\
7--71 & 3.2 $\times$ 10$^{-4}$ & 1.3 $\times$ 10$^{-2}$ & 8.723 $\pm$ 0.003 & 8.723 $\pm$ 0.002  Bo78   &     &     & H   \\ % (Te=2E+07 Ne=1E+20) Int=4.41e-02
7--68 & 1.1 $\times$ 10$^{-3}$ & 2.1 $\times$ 10$^{-2}$ & 8.731 $\pm$ 0.003 & 8.731 $\pm$ 0.002  Bo78   &   B78   &     &  H  \\ % Int=1.25e-03
9--95 & 6.6 $\times$ 10$^{-4}$ & 4.0 $\times$ 10$^{-2}$ & 8.752 $\pm$ 0.003 & 8.752 $\pm$ 0.002  Bo78   &   B78  &     &  H  \\ %
8--69 & 7.5 $\times$ 10$^{-4}$ & 2.6 $\times$ 10$^{-2}$ & 8.763 $\pm$ 0.003 & 8.763 $\pm$ 0.002   Bo78  &   B78  &     &   H  \\ %
9--86 & 2.0 $\times$ 10$^{-4}$ & 9.5 $\times$ 10$^{-3}$ & 8.764 $\pm$ 0.003 & ''    &     &     &   H \\ % (Te=2E+07 Ne=1E+20) Int=1.76e-02
5--56 & 0.20 & 6.9 $\times$ 10$^{-2}$ & 8.814 $\pm$ 0.002 & 8.814 $\pm$ 0.002   Bo78 (8.815 W98)  &  B78  &     &    \\  % Int=2.51e-01
5--48 & 8.4 $\times$ 10$^{-2}$ & 1.5 $\times$ 10$^{-2}$ & 8.906 $\pm$ 0.002 & 8.906 $\pm$ 0.001 W98 (?)   Bo78 (8.908) &   &     &    \\  % Int=1.01e-01
%1--25 & 6.5 10$^{-3}$ & 1.1 10$^{-2}$ & 10.506 $\pm$ 0.003 &      &     &      &   \\ 
%1--22 & 2.8 10$^{-3}$ & 1.2 10$^{-2}$ & 10.677 $\pm$ 0.006 &      &     &      &   \\ 
%3-35 & 3.2 10$^{-4}$ & 7.3 10$^{-3}$ & 10.794 $\pm$ 0.002 &      &     &      &   \\ 
\par
4--38 & 4.6 $\times$ 10$^{-3}$ & 0.15 & 10.813 $\pm$ 0.002 & 10.813 $\pm$ 0.003  Bo78   &     &     &  H  \\ % (Te=2E+07 Ne=1E+20) Int=5.77e-01
\par
%3-30 & 7.7 $\times$ 10$^{-4}$ & 2.4 $\times$ 10$^{-2}$ & 10.826 $\pm$ 0.006 &      &     &      &   \\ 
%4--37 & 1.9 $\times$ 10$^{-3}$ & 2.2 $\times$ 10$^{-2}$ & 10.827 $\pm$ 0.001 &      &     &      &   \\ 
\par
2--27 & 4.6 $\times$ 10$^{-3}$ & 4.9 $\times$ 10$^{-2}$ & 10.896 $\pm$ 0.007 & 10.903  &     &     &  H  \\
4--35 & 5.2 $\times$ 10$^{-3}$ & 0.12 & 10.903 $\pm$ 0.002 & 10.903 $\pm$ 0.003  Bo78 (10.907 S80)  &  F79   &   Bo78  &  H  \\ % (Te=2E+07 Ne=1E+20) Int=3.06e-01
4--33 & 2.1 $\times$ 10$^{-3}$ & 6.9 $\times$ 10$^{-2}$ & 10.906 $\pm$ 0.006 & 10.903  &     &     &  H  \\
\par
%3-28 & 2.3 10$^{-3}$ & 4.8 10$^{-2}$ & 10.910 $\pm$ 0.006 &      &     &      &   \\ 
\par
4--31 & 6.0 $\times$ 10$^{-3}$ & 0.12 & 10.927 $\pm$ 0.001 & 10.927 $\pm$ 0.003  Bo78  (10.925 S80, bl ?)  &  B78, F79   &     &   H    \\  % (Te=2E+07 Ne=1E+20) Int=4.70e-01
\par
%3-27 & 1.4 10$^{-3}$ & 1.5 10$^{-2}$ & 10.933 $\pm$ 0.006 &      &     &      &   \\ 
3--26 & 9.6 $\times$ 10$^{-3}$ & 0.10 & 10.935 $\pm$ 0.001 & 10.935 $\pm$ 0.003  Bo78    &     B78   &  Bo78, S80   &  H  \\  % (Te=2E+07 Ne=1E+20) Int=3.06e-01
\par
%4-30 & 2.8 10$^{-4}$ & 8.8 10$^{-3}$ & 10.936 $\pm$ 0.006 &      &     &      &   \\ 
%2-23 & 1.5 10$^{-3}$ & 1.3 10$^{-2}$ & 10.977 $\pm$ 0.007 &      &     &      &   \\ 
\par
1--15 & 1.0 & 0.29 & 10.980 $\pm$ 0.001 & 10.980 $\pm$ 0.003  Bo78, S80    &  B78, F79  &  FH75, S80   &    \\  % Int=1.00e+00
\par
%3-23 & 3.0 10$^{-3}$ & 2.7 10$^{-2}$ & 11.014 $\pm$ 0.006 &      &     &      &   \\ 
\par
1--13 & 0.64 & 0.28 & 11.018 $\pm$ 0.001 & 11.018 $\pm$ 0.003   Bo78, S80
(bl \ion{Fe}{xviii}, \xxiv)   &   B78, F79  &   S80   &    \\  % Int=6.54e-01
\par
%5--44 & 4.7 10$^{-3}$ & 1.8 10$^{-2}$ & 11.069 $\pm$ 0.006 &      &     &      &   \\ 
\par
5--37 & 1.1 $\times$ 10$^{-2}$ & 0.13 & 11.166 $\pm$ 0.001 & 11.166 $\pm$ 0.003  Bo78   &   B78 &   Bo78, S80   &  H  \\  % (Te=2E+07 Ne=1E+20) Int=4.35e-01
\par
%3-20 & 1.4 10$^{-2}$ & 4.1 10$^{-3}$ & 11.244 $\pm$ 0.003 &      &     &      &   \\ 
\par
5--35 & 2.3 $\times$ 10$^{-3}$ & 5.4 $\times$ 10$^{-2}$ & 11.247 $\pm$ 0.003 & 11.249 $\pm$ 0.003   Bo78   &     &    Bo78   &  H  \\ % (Te=2E+07 Ne=1E+20) Int=1.24e-01
%5--33 & 2.4 10$^{-4}$ & 7.6 10$^{-3}$ & 11.251 $\pm$ 0.006 &      &     &      &   \\ 
\par
5--30 & 1.4 $\times$ 10$^{-3}$ & 4.2 $\times$ 10$^{-2}$ & 11.282 $\pm$ 0.006 & 11.280 $\pm$ 0.003  Bo78  (bl ?)  &  &     &  H  \\
2--17 & 7.5 $\times$ 10$^{-2}$ & 0.28 & 11.299 $\pm$ 0.004 & 11.299 $\pm$ 0.003  Bo78 (11.294 S80)  &  B78, F79, S80, B85  &  Bo78   &    \\  % (Te=2E+07 Ne=1E+20) Int=1.09e+00
\par
3--18 & 0.13 & 0.59 & 11.337 $\pm$ 0.004 & 11.333 $\pm$ 0.003   Bo78 (bl \xviii)  &  F79  &   Bo78, B78    &     \\ % (Te=2E+07 Ne=1E+20) Int=2.35e+00 % NOT 11.325
\par
    &       &      &                  &   & &  B85, S80    &     \\
3--17 & 5.4 $\times$ 10$^{-2}$ & 0.21 & 11.339 $\pm$ 0.003 & (bl 11.333)    &     &     &   H \\  % (Te=2E+07 Ne=1E+20) Int=7.88e-01
\par
%5-27 & 2.2 10$^{-3}$ & 2.3 10$^{-2}$ & 11.398 $\pm$ 0.007 &      &     &      &   \\ 
%5-26 & 9.0 10$^{-4}$ & 9.6 10$^{-3}$ & 11.400 $\pm$ 0.001 &      &     &      &   \\ 
%8-45 & 6.0 10$^{-4}$ & 2.0 10$^{-2}$ & 11.419 $\pm$ 0.006 &      &     &      &   \\ 
%7-43 & 1.2 10$^{-3}$ & 4.2 10$^{-2}$ & 11.430 $\pm$ 0.007 &      &     &      &   \\ 
\par
%7-42 & 3.2 10$^{-3}$ & 8.3 10$^{-2}$ & 11.433 $\pm$ 0.007 &      &     &      &   \\ 
%7-41 & 5.5 10$^{-4}$ & 2.1 10$^{-2}$ & 11.434 $\pm$ 0.003 &      &     &      &   \\ 
\par
4--19 & 0.23 & 1.0 & 11.442 $\pm$ 0.003 & 11.442 $\pm$ 0.003  Bo78 (bl \xviii)  & F74, B78, S80, B85    &   Bo78   &    \\  % Int=1.69e-01
\par
%3-16 & 3.0 10$^{-3}$ & 1.0 10$^{-2}$ & 11.451 $\pm$ 0.003 &      &     &      &   \\ 
4--18 & 4.2 $\times$ 10$^{-2}$ & 0.19 & 11.457 $\pm$ 0.004 & 11.459 $\pm$ 0.003  Bo78 (bl ?)     &     &     &  H   \\ % (Te=2E+07 Ne=1E+20) Int=7.58e-01
\par
%4--17 & 3.6 10$^{-3}$ & 1.4 10$^{-2}$ & 11.459 $\pm$ 0.003 &      &     &      &   \\ 
\par
6--36 & 2.3 $\times$ 10$^{-2}$ & 0.17 & 11.485 $\pm$ 0.005 & 11.485 $\pm$ 0.003 Bo78 (11.495 S80)  &     &  &  H  \\ % (Te=2E+07 Ne=1E+20) Int=1.95e-01  % Int=2.09e-02
\par
%5-23 & 2.1 10$^{-3}$ & 1.9 10$^{-2}$ & 11.487 $\pm$ 0.007 &      &     &      &   \\ 
\par
%8-42 & 1.5 10$^{-3}$ & 3.9 10$^{-2}$ & 11.491 $\pm$ 0.007 &      &     &      &   \\ 
\par
7--39 & 2.8 $\times$ 10$^{-3}$ & 0.14 & 11.493 $\pm$ 0.005 & 11.493 $\pm$ 0.003 Bo78   &  B85  & B78    &   H  \\ % (Te=2E+07 Ne=1E+20) Int=2.53e-01
8--41 & 3.1 $\times$ 10$^{-3}$ & 0.12 & 11.493 $\pm$ 0.003 & 11.493 $\pm$ 0.003 Bo78    &     &     &  H   \\ % (Te=2E+07 Ne=1E+20) Int=1.92e-01
\par
8--40 & 3.7 $\times$ 10$^{-3}$ & 0.34 & 11.519 $\pm$ 0.003 & 11.519 $\pm$ 0.003  Bo78, S80  &  FH75, F79, B85   &   B78  &  H  \\ % (Te=2E+07 Ne=1E+20) Int=4.74e-01
\par
%8-39 & 5.6 10$^{-4}$ & 2.8 10$^{-2}$ & 11.552 $\pm$ 0.005 &      &     &      &   \\ 
\par
4--16 & 2.3 $\times$ 10$^{-3}$ & 7.6 $\times$ 10$^{-3}$ & 11.574 $\pm$ 0.003 &      &     &      &   \\ 
\par
%7--36 & 4.9 10$^{-3}$ & 3.6 10$^{-2}$ & 11.580 $\pm$ 0.005 &      &     &      &   \\ 
\par
9--45 & 9.5 $\times$ 10$^{-3}$ & 0.31 & 11.594 $\pm$ 0.006 & ?  11.594 $\pm$ 0.003 Bo78  &   Bo78, B78, B85, S80  &    & H   \\  
\hline
7--34 & 1.1 $\times$ 10$^{-2}$ & 0.18 & 11.614 $\pm$ 0.005 & 11.614 $\pm$ 0.003   Bo78, S80    &   F79, B85  &   Bo78, S80  &   H   \\ % (Te=2E+07 Ne=1E+20) Int=2.95e-01
%4--14 & 3.4 10$^{-3}$ & 9.6 10$^{-3}$ & 11.623 $\pm$ 0.007 &      &     &      &   \\ 
\par
%3-12 & 8.3 10$^{-3}$ & 1.5 10$^{-3}$ & 11.633 $\pm$ 0.001 &      &     &      &   \\ 
%9-42 & 4.3 10$^{-4}$ & 1.1 10$^{-2}$ & 11.669 $\pm$ 0.007 &      &     &      &   \\ 
\par
9--41 & 2.1 $\times$ 10$^{-3}$ & 8.2 $\times$ 10$^{-2}$ & 11.671 $\pm$ 0.003 & 11.669 $\pm$ 0.003 Bo78 (11.672 B85) &  B85   &     &  H  \\ % (Te=2E+07 Ne=1E+20) Int=1.21e-01
\par
%8-34 & 1.4 10$^{-3}$ & 2.4 10$^{-2}$ & 11.674 $\pm$ 0.005 &      &     &      &   \\ 
%2-11 & 1.4 10$^{-2}$ & 1.7 10$^{-2}$ & 11.675 $\pm$ 0.004 &      &     &      &   \\ 
%7--29 & 9.9 10$^{-4}$ & 1.2 10$^{-2}$ & 11.688 $\pm$ 0.008 &      &     &      &   \\ 
8--32 & 7.7 $\times$ 10$^{-3}$ & 0.23 & 11.692 $\pm$ 0.002 & 11.692 $\pm$ 0.003  Bo78 (11.694 S80)   &  F79, S80, B85    &  Bo78   &  H  \\ % (Te=2E+07 Ne=1E+20) Int=5.38e-01
\par
%9-40 & 2.5 10$^{-4}$ & 2.3 10$^{-2}$ & 11.698 $\pm$ 0.003 &      &     &      &   \\ 
\par
3--11 & 4.1 $\times$ 10$^{-2}$ & 5.1 $\times$ 10$^{-2}$ & 11.718 $\pm$ 0.003 & ?  11.718 $\pm$ 0.003  Bo78   & F79    &     &  \\ % (Te=2E+07 Ne=1E+20) Int=1.75e-01
\par
%9-39 & 5.8 10$^{-4}$ & 2.9 10$^{-2}$ & 11.732 $\pm$ 0.005 &      &     &      &   \\ 
5--20 & 2.2 & 0.67 & 11.737 $\pm$ 0.003 & 11.737 $\pm$ 0.003 Bo78  (11.742 S80)   &  N67, Bo78, B78, B85   & S80    &    \\  % Int=2.02e+00
8--29 & 7.1 $\times$ 10$^{-3}$ & 8.7 $\times$ 10$^{-2}$ & 11.749 $\pm$ 0.008 & 11.748 $\pm$ 0.003  Bo78 (bl ?)    &     &     &  H  \\ % (Te=2E+07 Ne=1E+20) Int=2.42e-01
\par
%7--24 & 1.3 $\times$ 10$^{-3}$ & 6.2 10$^{-2}$ & 11.835 $\pm$ 0.005 &      &     &      &   \\ 
4--11 & 7.1 $\times$ 10$^{-2}$ & 8.9 $\times$ 10$^{-2}$ & 11.846 $\pm$ 0.003 & weak     &     &     &  H  \\ % 11.846 $\pm$ 0.003  (bl)  Bo78 (Te=2E+07 Ne=1E+20) Int=3.01e-01
\par
%9-34 & 1.6 10$^{-3}$ & 2.6 10$^{-2}$ & 11.858 $\pm$ 0.005 &      &     &      &   \\ 
10--46 & 7.6 $\times$ 10$^{-2}$ & 8.4 $\times$ 10$^{-2}$ & 11.874 $\pm$ 0.008 & weak    &     &     &     \\ % (Te=2E+07 Ne=1E+20) Int=2.01e-01
\par
%9-32 & 2.3 10$^{-4}$ & 7.1 10$^{-3}$ & 11.876 $\pm$ 0.002 &      &     &      &   \\ 
%6-22 & 1.0 10$^{-2}$ & 4.6 10$^{-2}$ & 11.891 $\pm$ 0.008 &      &     &      &   \\ 
8--24 & 2.2 $\times$ 10$^{-3}$ & 0.11 & 11.898 $\pm$ 0.005 & 11.898 $\pm$ 0.003  Bo78 (bl ?)  &     &  S80  &  H  \\ % (Te=2E+07 Ne=1E+20) Int=3.99e-01
\par
%9-29 & 3.4 10$^{-3}$ & 4.2 10$^{-2}$ & 11.935 $\pm$ 0.008 &      &     &      &   \\ 
%7-22 & 5.8 10$^{-3}$ & 2.6 10$^{-2}$ & 11.993 $\pm$ 0.008 &      &     &      &   \\ 
\par
7--21 & 7.4 $\times$ 10$^{-3}$ & 5.0 $\times$ 10$^{-2}$ & 12.018 $\pm$ 0.008 & ?  12.027 $\pm$ 0.003  Bo78   &     &     &   H   \\
9--25 & 3.5 $\times$ 10$^{-2}$ & 5.9 $\times$ 10$^{-2}$ & 12.027 $\pm$ 0.005 & ?  12.027 $\pm$ 0.003  Bo78   &     &     &   H   \\ % (Te=2E+07 Ne=1E+20) Int=3.41e-01
8--22 & 1.5 $\times$ 10$^{-2}$ & 6.6 $\times$ 10$^{-2}$ & 12.057 $\pm$ 0.008 & ?  12.053 $\pm$ 0.003  Bo78    &     &     &   H  \\ % (Te=2E+07 Ne=1E+20) Int=2.21e-01
%9-24 & 6.3 10$^{-4}$ & 3.1 10$^{-2}$ & 12.089 $\pm$ 0.005 &      &     &      &   \\ 
\par
5--12 & 1.2 & 0.23 & 12.161 $\pm$ 0.002 & 12.161 $\pm$ 0.005   B02  &     &   Bo78    &    \\ % not  12.138 Int=1.02e+00
%10-25 & 1.5 10$^{-2}$ & 2.5 10$^{-2}$ & 12.351 $\pm$ 0.006 &      &     &      &   \\ 
7--16 & 3.7 $\times$ 10$^{-2}$ & 0.12 & 12.369 $\pm$ 0.004 & ? bl 12.380 $\pm$ 0.003  Bo78   &     &     &   H \\ % Int=2.55e-02
\par
%7--14 & 2.5 10$^{-2}$ & 7.1 10$^{-2}$ & 12.425 $\pm$ 0.008 &      &     &      &   \\ 
8--16 & 4.9 10$^{-2}$ & 0.16 & 12.437 $\pm$ 0.004 & ? 12.438 $\pm$ 0.003  Bo78   &     &     &  H  \\ % Int=3.46e-02 % (Te=2E+07 Ne=1E+20) Int=6.02e-01
\par
%8-15 & 8.1 10$^{-3}$ & 2.4 10$^{-3}$ & 12.444 $\pm$ 0.002 &      &     &      &   \\ 
%8-13 & 1.3 10$^{-2}$ & 5.7 10$^{-3}$ & 12.493 $\pm$ 0.002 &      &     &      &   \\ 
%9-16 & 1.0 10$^{-2}$ & 3.5 10$^{-2}$ & 12.646 $\pm$ 0.004 &      &     &      &   \\ 
%9-15 & 3.0 10$^{-2}$ & 8.9 10$^{-3}$ & 12.653 $\pm$ 0.002 &      &     &      &   \\ 
%9-13 & 8.3 10$^{-3}$ & 3.6 10$^{-3}$ & 12.703 $\pm$ 0.002 &      &     &      &   \\ 
%11-51 & 1.7 10$^{-2}$ & 3.9 10$^{-2}$ & 32.010 $\pm$ 0.072 &      &     &      &   \\ 
%11-50 & 2.0 10$^{-2}$ & 1.3 10$^{-2}$ & 32.120 $\pm$ 0.052 &      &     &      &   \\ 
%12-52 & 2.5 10$^{-2}$ & 6.1 10$^{-3}$ & 32.587 $\pm$ 0.064 &      &     &      &   \\ 
%13-56 & 1.6 10$^{-2}$ & 5.5 10$^{-3}$ & 33.091 $\pm$ 0.033 &      &     &      &   \\ 
%13-54 & 3.7 10$^{-3}$ & 1.3 10$^{-2}$ & 33.345 $\pm$ 0.033 &      &     &      &   \\ 
%15-56 & 4.3 10$^{-2}$ & 1.5 10$^{-2}$ & 33.439 $\pm$ 0.034 &      &     &      &   \\ 
%31-82 & 8.7 10$^{-4}$ & 2.2 10$^{-2}$ & 33.676 $\pm$ 0.068 &      &     &      &   \\ 
%16-55 & 8.8 10$^{-3}$ & 2.8 10$^{-2}$ & 33.684 $\pm$ 0.045 &      &     &      &   \\ 
%13-48 & 2.7 10$^{-2}$ & 4.9 10$^{-3}$ & 34.426 $\pm$ 0.036 &      &     &      &   \\ 
%18-58 & 1.6 10$^{-2}$ & 1.6 10$^{-2}$ & 34.450 $\pm$ 0.095 &      &     &      &   \\ 
%19-59 & 2.0 10$^{-2}$ & 2.2 10$^{-2}$ & 34.565 $\pm$ 0.083 &      &     &      &   \\ 
%15-48 & 5.0 10$^{-2}$ & 9.2 10$^{-3}$ & 34.803 $\pm$ 0.036 &      &     &      &   \\ 
%20-60 & 9.9 10$^{-2}$ & 2.6 10$^{-2}$ & 35.250 $\pm$ 0.087 &      &     &      &   \\ 
%3-9 & 1.8 10$^{-2}$ & 0.22 & 121.201 $\pm$ 0.076 &      &     &      &   \\ 
\par
1--5 & 82. & 12. & 132.906 $\pm$ 0.018 & 132.906 $\pm$ 0.005 SR95  (bl) (132.84 LP80)   &  K74 (132.83)  &     &    \\ % Int=5.21e+00 % (Te=2E+07 Ne=1E+20) Int=8.68e-01
\par
4--9 & 0.18 & 2.2 & 136.531 $\pm$ 0.095 & 136.53 $\pm$ 0.03 LP80  (136.67 B82)   & LP80    &     &  H  \\ % Int=1.09e-02
3--8 & 0.13 & 3.4 & 144.389 $\pm$ 0.106 & 144.36 $\pm$ 0.03  LP80 (144.30 B82)  &  LP80   &     &  H  \\ % NOT 144.390  Int=6.06e-03
\par
2--7 & 0.28 & 2.7 & 147.254 $\pm$ 0.281 & 147.24 $\pm$ 0.03  LP80  (147.27 B82)  &  LP80   &     &  H  \\ % 147.271 Int=1.52e-02
5--10 & 0.25 & 1.3 & 149.211 $\pm$ 0.200 & 149.22 $\pm$ 0.03  LP80  (149.37 B82)   & LP80    &     &  H  \\ % Int=1.57e-02
\par
3--7 & 0.18 & 1.7 & 154.303 $\pm$ 0.121 & 154.27 $\pm$ 0.03  LP80    &  LP80   &     &  H  \\ % 154.304
4--8 & 0.18 & 4.7 & 166.686 $\pm$ 0.142 & 166.74 $\pm$ 0.03  LP80  (166.71 B82)   &  LP80   &     &  H  \\ % not 166.689
3--6 & 8.5 $\times$ 10$^{-2}$ & 2.1 & 173.318 $\pm$ 0.153 & 173.31 $\pm$ 0.03  LP80  (173.47 B82)   &  LP80   &     &  H  \\
4--7 & 0.19 & 1.8 & 180.040 $\pm$ 0.165 & 180.10 $\pm$ 0.06  LP80  (180.1 B82)  &  LP80   &     &  H  \\ % 180.044
\par
%19-38 & 9.2 10$^{-4}$ & 2.9 10$^{-2}$ & 196.618 $\pm$ 1.534 &      &     &      &   \\ 
\par
5--9 & 0.17 & 2.0 & 221.342 $\pm$ 0.294 & 221.33 $\pm$ 0.06  LP80  (221.50 B82)  & LP80    &     &  H  \\ % 221.553
1--3 & 6.2 & 9.9 $\times$ 10$^{-2}$ & 263.765 $\pm$ 0.007 & 263.765 $\pm$ 0.005 SR95 &   W75 (263.76)  &     &    \\ %
\par
%5-8 & 9.4 10$^{-3}$ & 0.24 & 313.195 $\pm$ 0.587 &      &     &      &   \\ 
%5-6 & 1.6 10$^{-4}$ & 3.9 10$^{-3}$ & 490.942 $\pm$ 1.442 &      &     &      &   \\ 
%11-16 & 1.7 10$^{-3}$ & 5.6 10$^{-3}$ & 503.018 $\pm$ 9.921 &      &     &      &   \\ 
%64-97 & 5.5 10$^{-5}$ & 5.8 10$^{-5}$ & 504.304 $\pm$ 24.211 &      &     &      &   \\ 
%13-20 & 4.1 10$^{-5}$ & 1.2 10$^{-5}$ & 509.061 $\pm$ 7.657 &      &     &      &   \\ 
%11-15 & 5.7 10$^{-5}$ & 1.7 10$^{-5}$ & 514.668 $\pm$ 7.826 &      &     &      &   \\ 
%25-37 & 2.0 10$^{-6}$ & 2.3 10$^{-5}$ & 528.541 $\pm$ 10.943 &      &     &      &   \\ 
%21-27 & 1.0 10$^{-6}$ & 1.1 10$^{-5}$ & 563.180 $\pm$ 30.026 &      &     &      &   \\ 
%24-35 & 8.2 10$^{-7}$ & 1.9 10$^{-5}$ & 598.086 $\pm$ 17.366 &      &     &      &   \\ 
%15-20 & 5.4 10$^{-5}$ & 1.6 10$^{-5}$ & 606.061 $\pm$ 10.823 &      &     &      &   \\ 
%11-13 & 6.6 10$^{-5}$ & 2.8 10$^{-5}$ & 614.024 $\pm$ 11.106 &      &     &      &   \\ 
%11-14 & 4.8 10$^{-4}$ & 1.3 10$^{-3}$ & 617.181 $\pm$ 25.560 &      &     &      &   \\ 
%22-26 & 1.8 10$^{-6}$ & 1.9 10$^{-5}$ & 631.481 $\pm$ 23.053 &      &     &      &   \\ 
%24-31 & 8.2 10$^{-7}$ & 1.7 10$^{-5}$ & 680.735 $\pm$ 18.045 &      &     &      &   \\ 
%73-97 & 1.8 10$^{-5}$ & 2.0 10$^{-5}$ & 710.778 $\pm$ 47.168 &      &     &      &   \\ 
%12-15 & 2.5 10$^{-5}$ & 7.4 10$^{-6}$ & 756.430 $\pm$ 11.273 &      &     &      &   \\ 
%23-29 & 1.1 10$^{-6}$ & 1.4 10$^{-5}$ & 800.352 $\pm$ 59.310 &      &     &      &   \\ 
%2-4 & 1.7 10$^{-4}$ & 1.9 10$^{-9}$ & 808.551 $\pm$ 5.325 &      &     &      &   \\ 
%31-38 & 6.0 10$^{-5}$ & 1.9 10$^{-3}$ & 1036.27 $\pm$ 31.24 &      &     &      &   \\ 
\par
3--4 & 2.5 & 2.7 $\times$ 10$^{-5}$ & 1079.41 $\pm$ 0.23 & 1079.41 $\pm$ 0.03 F00 (1079.3 $\pm$ 0.3 H80)  &  H80, F00   &     &    \\ 
 \hline
\end{tabular}
\normalsize
%\label{tab:id}
%Table~\ref{tab:id}
\end{table}